The Relationship Between Maternal Obesity With Pregnancy-associated Hypothyroidism, Fetal Health, and Pregnancy Outcomes



Lourance Al Hadid¹ (0), Omymah Z. Al-Rajabi² (0), Marwa Al Barmawi³ (0), Ahmad Yahya AL-Sagarat^{4*} (0)

1. Associate Professor, Faculty of Nursing, Al-Balqa Applied University, Salt, Jordan.

2. Instructor, Rufaida Al-Aslamiah College of Nursing, Al-Balqa Applied University, Salt, Jordan.

3. Associate Professor, Faculty of Nursing, Al-Zaytoonah University of Jordan (ZUJ), Amman, Jordan.

4. Associate Professor of Mental Health Nursing, Department of Community and Mental Health Nursing, Faculty of Nursing, Mutah University, AL-Karak, Jordan.



Citation Al Hadid L, Al-Rajabi OZ, Al Barmawi M, AL-Sagarat AY. The Relationship Between Maternal Obesity With Pregnancyassociated Hypothyroidism, Fetal Health, and Pregnancy Outcomes. J Holist Nurs Midwifery. 2023; 33(3):183-192.https://doi. org/10.32598/jhnm.33.3.2400

Running Title Obesity, Hypothyroidism, and Pregnancy Outcomes

doj https://doi.org/10.32598/jhnm.33.3.2400

Article info: Received: 17/02/2022 Accepted: 5/03/2023 Available Online: 01/07/2023

ABSTRACT

Introduction: The prevalence of obesity is increasing worldwide, especially among pregnant women. Maternal obesity is a serious risk factor during pregnancy associated with pregnancy-related complications for the mother and her child.

Objective: This study aimed to estimate the prevalence of Jordanian pregnant women who are overweight or obese. The relationship between high body mass index and selected obstetrical conditions was also investigated.

Materials and Methods: A descriptive, correlational, cross-sectional design was used on a purposive convenience sample of 411 pregnant women in Jordan. Results were obtained through structured interviews, antenatal routine lab tests, physical measurements (height and weight), body mass index (BMI), thyroid function tests (free thyroxine [FT4] and 2e [TSH]), and Edinburgh postnatal depression scale. Descriptive statistics, including means, standard deviation, and inferential statistics, such as Pearson correlation, t-test, and ANOVA, were used to describe and examine the relationship among the study variables.

Results: Based on the results, 58.4% of the participants had one to four pregnancies, and 54.5% had at least one abortion. Based on their weight before pregnancy, there was a high frequency of overweight (29.7%) and obesity (25%) among pregnant women. High maternal BMI for the mother is correlated significantly with pregnancy complications and offspring neurocognitive impairments. Additionally, maternal obesity is a risk factor for hypothyroidism. BMI was correlated with hypothyroidism (r=0.141, P=0.004), fetal distress (r=0.217, P=0.0001), postnatal depression (r=0.161, P=0.0001), and preterm labor (r=0.115, P=0.020). The thyroid function tests and thyroid stimulating hormone levels were correlated with maternal and fetal conditions, such as infertility, fetal death, hemorrhage, and cerebral palsy among children.

Keywords:

Maternal obesity, Hypothyroidism, Pregnancy outcomes **Conclusion:** High BMI associated with old age is a critical factor related to many maternal and infant health conditions. Repeated pregnancy failures, fetal health conditions, and maternal complications, including hypothyroidism, can be associated with maternal obesity. Healthcare providers should raise awareness among mothers and care providers on the serious conditions associated with obesity.

* Corresponding Author:

Ahmad Yahya AL-Sagarat, Associate Professor.

Address: Department of Community and Mental Health Nursing, Faculty of Nursing, Mutah University, AL-Karak, Jordan. Tel: +96 (27) 72038527

.....

E-mail: sagarat74@mutah.edu.jo, sagarat4@hotmail.com



Highlights

- There is a high prevalence of obesity in pregnant women.
- There is an association between obesity and pregnancy-related complications.
- High body mass index (BMI) among pregnant women affects the well-being of the offspring.
- Maternal obesity is a risk factor for hypothyroidism.

Plain Language Summary

Mother weight and thyroid function are important factors that interfere with maternal, fetal, and born-child wellbeing. Findings in this study indicated that the greater the mother's weight, the higher the opportunity for the mother and her child to experience health problems, such as gestational diabetes, postpartum hemorrhage, fetal distress and death, and cerebral palsy for the born child. In addition, decreased thyroid function affects the outcome of pregnancy, which might include the mother's and her child's health.

Introduction

lobally, the prevalence of obesity is increas-

ing, especially among women. Over 50% of women were obese in 138 countries [1]. Maternal obesity is a serious risk factor during pregnancy; obesity in the United States affects 0.29% of women before becoming preg-

nant [2]. Obesity among pregnant women increases the prevalence of pregnancy-related complications for both mother and child [3]. Defining obesity and overweight according to body mass index (BMI) is a widely accepted [4]. High BMI can be defined as overweight (BMI>24.9 kg/m²) or obesity (BMI >29.9 kg/m²) [4]. Obesity and overweight have been documented to affect the mother and her offspring. These complications include prematurity, stillbirth, preterm labor, neural defects, congenital anomalies, miscarriage, postpartum hemorrhage, and fetal distress [5-8]. Additionally, obesity has been associated with reproductive system complications even before conception. Infertility, sub-fertility, and irregular cycles were linked to high BMI [9, 10].

Furthermore, a population-based birth cohort study on 2,379 children at early ages in Spain and Greece revealed an association between prepregnancy obesity and neuropsychological impairment with the learning disabilities of the children [11]. Obesity is a risk factor for thyroid dysfunction during pregnancy. Hypothyroidism affects approximately 2%-2.5% of pregnant women; 0.2%-0.5% of pregnancies with overt hypothyroidism [12]. While this condition increases the risk for pregnancy-related complications, subclinical hypothyroidism (increased thyroid stimulating hormone [TSH] and normal free thyroxine [FT4]) and isolated hypothyroxinemia (normal TSH, decreased FT4) were also found to increase the risks of preeclampsia, placental abruption, spontaneous abortion, postpartum hemorrhage, preterm birth, intrauterine growth retardation, fetal distress, stillbirth, low birth weight, congenital malformations, and offspring neuropsychological impairments [13, 14].

Furthermore, age is a risk factor for obesity. It plays a negative role in thyroid dysfunction tests. A prospective study on 1035 Dutch Caucasian pregnant women revealed a negative association between obesity, primiparity at a younger age, and FT4; higher TSH, pregnancy complications, and offspring impairments were documented with hypothyroidism [15]. Similar findings has been reported in Poland [6].

Thyroxine deficiency (i.e. decreased FT4) with normal TSH in pregnancy is a risk for developing hypothyroxinemia. This condition occurs in the presence of low maternal iodine intake and high consumption by the fetus. A cluster analysis study on 306 otherwise healthy Mexican pregnant women found that obese women with high BMI are at greater risk of developing hypothyroxinemia [16]. Hypothyroxinemia was documented to negatively impact psychological and neurological development, attention deficit and hyperactivity disorders, and autism among children [17, 18]. Pregnancy outcomes in this study refer to any condition affecting the mother or child during pregnancy and labor.

In Jordan, this issue has not been adequately addressed in empirical studies. Therefore, this study aimed to estimate the prevalence of Jordanian pregnant women who are overweight or obese. This study also investigated the relationship between increased BMI with hypothyroidism and selected obstetrical outcomes.

Materials and Methods

A descriptive, correlational, cross-sectional design was used to assess the frequency of maternal obesity, hypothyroidism, and other obstetrical negative outcomes and their correlation with increased BMI.

The target population presents all pregnant women aged 17-50 years old. The inclusion criteria were as follows: Any pregnant woman with no diabetes or hypertension or previous maternal complications (such as preeclampsia and miscarriage). Participant information was obtained from the medical records. The study sample was pregnant women selected from four mother and child health centers. They were purposively selected as they provided services to the most populated cities in Jordan, Amman, and Zarga. Both cities represented approximately half of the Jordanian population. These cities have 2763000 females, representing about onequarter of the total population and more than half the female population in Jordan [19]. Participants should receive antenatal care in these public health centers and have a follow-up file. The total sample size required for the study was calculated based on a confidence level of 95%, a margin of error of 5%, and a population size, assuming inferential tests (Pearson correlation, t-test, and ANOVA). The sample size required to represent the population was found to be 384 pregnant women considering the main variables examined in this study (i.e. pregnancy trimester, BMI, thyroid function, pregnancy outcome) using the statistical software G^{*}Power version 3.1.9.4 (medium effect size 0.25, alpha 0.05, and power 0.95). This number is suitable for all planned tests in this studyy [20]. A total of 411 pregnant women were surveyed to improve representation and decrease margin error. In addition, the attrition rate of 20% and the level of testing (i.e. analysis of variance) were considered when calculating the sample size.

The study questionnaire consists of three parts. Part one was demographic data, which included age, hemoglobin level, number of previous pregnancies, and number of abortions. Part two comprised the physical assessment, which included height, weight, and BMI measurement. Part three of the questionnaire included the lab results (TSH, FT4) and history of complications (such as infertility, missed abortion, fetal death, and fetal distress), anthropometric measurements, which included weight and height that were taken for all participants in light clothing, and no shoes during the three trimesters. Weight before pregnancy was obtained from records and was measured using one digital scale to the nearest 0.1 kg.

All measurements and tests were taken during the interview. Height was measured using measuring tape on the wall with the participant standing without shoes, looking straight with heels together, and recorded to the nearest 0.5 cm. The reliability of the anthropometric measures was maintained using the same protocol by the same person using the same equipment. The digital weight scale was used according to manufacturer instructions and calibrated before data collection, and each time it was moved from the place. BMI was calculated as the ratio of weight (kilograms) to the square of height (meters). Obesity is categorized according to age-sex-specific percentiles of BMI using the international obesity task force (IOTF) for age growth charts [21]. BMI was documented before pregnancy and at the end of each trimester; this information was obtained from the medical records. The study significance and instruction for the structured interview were explained to the participants, who are eligible and agreed to take the antenatal lab tests, physical measurements (height and weight), BMI, and thyroid function tests (TSH and FT4). It was explained to the candidates that neither refusing to participate nor withdraw would produce any burden or influence on the care provided to them. Participation was entirely voluntary and anonymous. The data were treated confidentially using code for names and used only by the researchers for research purposes. Information about women's health during previous pregnancies was taken from the digital and paper medical records in the healthcare center, including complications associated with pregnancy to the mother and the born child (including fetal death, repeated abortions, gestational hypertension, placenta previa, postpartum depression, low birth weight, fetal distress, early delivery, cerebral palsy, and children with a learning difficulty). Postpartum depression was measured using the Edinburgh postnatal depression scale (EDPS), 10 screening questions representing symptoms common in women with depression and anxiety during pregnancy. The EDPS English version is a 4-point Likert scale with a total score range between 0 and 30. A score greater than 8 indicates a higher possibility of depression [22].

Data were collected using a standardized interview to fill in the study questionnaire and take physical measurements by assigned data collectors to maintain consistency. The data collection was conducted in the period from March 2019 to March 2020 when the lockdown took place because of COVID-19. Each interviewee was interviewed in person and in the vicinity of the healthcare center in a private room, which was provided by the center manager for the researcher.

All data analyses were performed using SPSS software, version 21 (SPSS, Chicago, IL, USA) software. Minimum values, maximum values, and frequency tables were explored to identify missing data and outliers.

Descriptive statistics, including Means±SD and percentages, were used as appropriate to present the demographic variables, physical measurements, and results of lab tests. Further, percentages were used for pregnancy and offspring complications. Inferential statistics were used to measure the relationship between thyroid and BMI values on pregnancy conditions, such as the Pearson correlation, t-test, and ANOVA according to testing groups.

Results

A total number of 411 pregnant women participated in this study. More than half of the participants (58.4%) had one to four pregnancies, and many participants (54.5%) had at least one abortion (Table 1). It is worth noting here that elective abortion is illegal in Jordan, except in case of expected physical harm to the pregnant. Therefore, all abortions reported in this study were based on medical conditions. Previous abortions were reported in more than 45.5% of the participants, with 25.8% reporting more than two abortions (n=104).

This study found that 51.6% of participants were overweight or obese (Table 2). Of the obese participants, 14.6% had class-1 obesity, 5.6% had class-2 obesity, and 1.7% had class-3 morbid obesity. Just under half of the women (41.6%) were anemic (hemoglobin <11 mg/dL), and the majority (58.4%) had the minimum acceptable hemoglobin during pregnancy, more than 11.0 mg/dL. The value for TSH in the first trimester was 0.1-2.5 mlU/L, 0.2-3 mlU/L for the second, and 0.3-3.5 mIU/L for the third trimester. The normal value is < 3.5 mlU/L [12]. The FT4 is at the lower limit in the first trimester (13.68 Pmol/L; reference range: 12.1-19.6 Pmol/L). Most thyroid function using TSH and FT4 (73%) results were within the normal serum levels accounting for trimester-related levels. High serum levels of TSH distributed across the three trimesters were present in 26% of participants. Low FT4 levels were detected in 12.6% of participants. The relationship between high BMI (>25 kg/m²) and thyroid dysfunction test showed no statistically significant correlation.

Table 3 presents the rate of pregnancy conditions among the participants. There are high percentages noticed in conditions like postpartum depression (24.6%), fetal death (15.6%), and infertility conditions (14.4%) among obese/overweight compared with non-obese women. The correlations between BMI and the pregnant and the offspring health outcomes were tested using the Pearson correlation coefficient. High BMI was associated with higher postnatal depression (r=0.161,

Table 1. Participants' demographic and reproductive characteristics (n=411)

Factor		Mean±SD/No. (%)	Range
Age (y)		29.65±6.32	17-47
Hemoglobin (mg/dL)		11.10±1.40	7.0-16.0
Pregnancy trimester	1 st 2 nd 3 rd	157(38.2) 77(18.7) 177(43.1)	
Number of pregnancies		4.1±1.1	1-4: 240(58.4) 5-8: 147(35.8) >8: 24(5.8)
Number of abortions		0.99±0.25	0: 224(54.5) 1: 81(19.7) 2-4: 97(23.6) >4: 9(2.2)

Variables	Category	Description	No. (%)	Mean±SD
BMI category (kg/m²)	Underweight	<18.5	5(1.2)	26.33±5.77
	Normal	18.5-24.9	194(47.2)	
	Overweight	25-29.9	122(29.7)	
	Obese class1	30-34.9	60(14.6)	
	Obese class 2	35-39.9	23(5.6)	
	Obese class 3	>40	7(1.7)	
Hemoglobin (mg/dL)		<11	41.6	11.10±1.87
TSH (mlU/L)	1 st trimester	0.1-2.5*	157(38.2)	2.872±.54
	2 nd trimester	0.2-3*	77(18.7)	2.74±.37
	3 rd trimester	0.3-3.5*	177(43.1)	2.45±.54
	Combined		2.67*	
Interpretation of serum levels of TSH	Low		6(1.5)	
	Normal		297(72.3)	
	High		107(26)	
FT4 (Pmol/L)	1 st trimester	12.1-19.6**	157(38.2)	13.68±1.98
	2 nd trimester	9.6-17**	77(18.7)	11.87±2.13
	3 rd trimester	8.4-15.6**	177(43.1)	12.17±2.08
	Combined		12.69**	
Interpretation of serum levels of FT4	Low		53(12.9)	
	Normal		341(83)	
	High		17(4.1)	

Table 2. Physical measurement among participants (n=411)

Abbreviations: FT4: Free thyroxine; FT3: Free triiodothyronine; TSH: Thyroid-stimulating hormone; BMI: Body mass index.

*Milli-international unit per liter, **Picomoles per litter.

P=0.001), preterm labor (r=0.115, P=0.02), and fetal distress (r=0.217, P=0.001).

A significant correlation was also between BMI>30 kg/ m2 and hypothyroidism (r=0.141, P=0.004). No statistically significant relationship was found between pregnancy conditions and high BMI (Table 4).

Thyroid function and BMI had statistically significant correlations with the categories of a range of pregnancy conditions, including fertility (P=0.005), fetus death (P=0.023), thyroid disease (P=0.0001), and cardiovascular disease (P=0.001) due to Table 5.

Discussion

This study showed a high prevalence of overweight and obese pregnant women in all age groups. Similar results were reported in the USA [2]. Another study reported that 45% of pregnant women from South-Asian countries were either overweight or obese during their pregnancy [7]. However, one study reported that 26% of Southern European women were overweight or obese [11]. Another study reported that 29% of Australian pregnant women were overweight [23].

Condition	No. (%)		
Condition	Yes	No	
GDM	20(4.1)	391(95.1)	
Infertility condition	59(14.4)	352(85.6)	
Fetal death	64(15.6)	347(84.4)	
Repeated abortions	2(0.5)	409(99.5)	
Gestational hypertension	4(1)	407(99)	
Placenta previa	4(1)	407(99)	
Postpartum depression	101(24.6)	310(75.4)	
Low birth weight	56(13.6)	355(86.4)	
Fetal distress	38(9.2)	373(90.8)	
Early delivery	45(10.9)	366(89.1)	
Cerebral palsy	36(8.8)	375(91.2)	
Children with a learning difficulty	7(1.7)	404(98.4)	

Table 3. Reported pregnancy-related conditions among women (n=411)

GDM: Gestational diabetes mellitus.

Findings in this study have shown that many obstetrical complications have been associated with obesity, such as postnatal depression, preterm labor, and fetal distress. Similarly, numerous reports in the literature indicate significant associations between obesity and obstetrical and offspring complications [4, 6-8]. Furthermore, many of these outcomes were associated with hypothyroidism [6]. This study also revealed the association between high BMI and the risk of hypothyroidism.

In addition, age and high BMI (>35 kg/m²) were associated with pregnancy complications, offspring malformation, and cognitive impairments. Individuals over 35 are at an increased risk of developing hypothyroidism with either increased TSH level or decreased FT4. Olderage pregnant women are even at more risk for other pregnancy complications such as gestational diabetes, repeated miscarriages, gestational hypertension, and preterm labor, and have negative outcomes for offspring such as children with learning difficulties [13, 23] otherwise, complicated cases before pregnancy, such as fertility problems [9, 10].

Table 4. Correlation between BMI and pregnancy outcomes

	Association	r**	Р
Pc BMI	Hypothyroidism	0.141	0.004
	Postnatal depression	0.161	0.0001
	Preterm labor	0.115	0.020
	Fetal distress	0.217	0.0001

Gestational diabetes, abortion, stillbirth, preeclampsia, premature, congenital anomaly, gestational hypertension, subfertility secondary infertility (delayed conception), and postpartum hemorrhage had no significant statistically significant correlation (P>0.05).

**The Pearson correlation.



Factor	Variables	Mean Square	df	P**
FT4 level	Fertility condition	0.286	2	0.005
	Fetus death	0.244	2	0.023
	Cerebral palsy children	0.060	2	0.006
	GDM	0.054	236	0.009
	Cardiovascular disease	0.056	236	0.0001
	Thyroid disease	0.051	236	0.0001
	Fertility condition	0.150	236	0.0001
	Fetus death	0.169	236	0.0001
TCU	PP hemorrhage	0.107	236	0.010
TSH	PP thyroiditis	0.203	236	0.0001
	PP depression	0.224	236	0.0001
	Low weight birth	0.140	236	0.0001
	History of fetal distress	0.102	236	0.0001
	Preterm labor	0.130	236	0.0001
	Cerebral palsy children	0.110	236	0.0001
	Fertility condition	0.139	4	0.036
	Fetal death	0.170	4	0.033
BMI category	PP Thyroiditis	0.567	4	0.0001
	Low weight birth	0.186	4	0.0001
	Cerebral palsy children	0.083	4	0.0001
	Fetus death	0.146	8	0.024
	Hemorrhage history	0.160	8	0.030
	Thyroiditis history	0.179	8	0.011
	Depression history	0.305	8	0.001
FT4 level & TSH	Low weight birth	0.082	8	0.003
	History of fetal distress	0.144	8	0.0001
	Preterm labor	0.104	8	0.009
	Cerebral palsy children	0.057	8	0.0001
	Fertility condition	0.149	41	0.0001
TSH & BMI	Fetus death	0.102	41	0.025
	Thyroiditis history	0.183	41	0.0001
	Depression history	0.150	41	0.015
	Low weight birth	0.122	41	0.0001
	History of fetal distress	0.059	41	0.006
	Cerebral palsy children	0.066	41	0.0001

Table 5. Differences in the correlation of thyroid function, body mass index, and pregnancy-associated conditions

Abbreviations: FT4: Free thyroxine; FT3: Free triiodothyronine; TSH: Thyroid-stimulating hormone; BMI: Body mass index; GDM: Gestional diabetes melitus.

**The t-test and ANOVA test.

With FT4 levels outside the normal range, offspring are likelier to have low birth weight, cerebral palsy, and other pregnancy complications [16]. We found an association between abnormal TSH/FT4 levels, antepartum hemorrhage (placenta previa), thyroid dysfunction symptoms, and postpartum depression. Similarly, preeclampsia hypertension, preeclamptic toxemia (PET), and eclampsia were 3.5 times, 2.8 times, and 4.5 times higher for overweight pregnant women than non-overweight pregnant women in India [14, 25]. In Pakistan, PET, in obese pregnant women was more than 15% higher compared to normal-weight pregnant women, eclampsia was three-fold greater in obese women, and gestational diabetes mellitus was higher at 23.3% as compared to 6.7% in non-obese [26].

Older age pregnant women are more likely to develop thyroid dysfunction, pregnancy complications, and congenital or neurocognitive fetal complications [15, 26, 27]. However, other studies have suggested that obesity, primiparity, and younger age had abnormal levels of FT4 at all trimesters, higher TSH, pregnancy complications, and offspring problems [27, 28].

A study in India found that higher maternal age and high BMI are risk factors for developing hypothyroidism during pregnancy among 1000 women at a rate of 9.2%, with 8.5% subclinical and only 0.7% having overt hypothyroidism [29]. Extreme obesity also results in high levels of thyroid-stimulating hormone due to abnormality of the hypothalamic-pituitary-thyroid axis [30], supported by this study's findings. As reported in the literature [31], obesity has been associated with both physiological and psychological problems among different groups of individuals, such as pregnant mothers [32]. Therefore, we recommend conducting studies that examine associations between maternal obesity, long term psychological health outcomes. We also recommend conducting larger-scale studies to determine obstetric and offspring conditions after pregnancy among obese women and the subsequent impact on future health conditions and pregnancies.

There were limitations that we experienced in this study. The first limitation is the limited budgeting and data collection settings. This study did not receive any funds, and the research carried out study steps independently. Therefore, it was impossible to include more numbers of women, which represents a second and important limitation. In addition, convenience sampling, which was limited in the place and the time, is another limitation that could be avoided in future more presentative studies. Finally, another limitation is the absence of regional or local reference levels of tests. Hence, we urge researchers to conduct future research on a much greater scale and include plans of defining reference values that apply to middle eastern populations.

The findings in this study documented many pregnancy-associated complications of obesity. Therefore, it is imperative to work on different aspects leading to increased safety of the pregnant woman and the pregnancy. BMI and hypothyroidism are risk factors that can be recognized and thus managed at a low cost. Screening of women, especially among women older than 35 years of age, is a proactive step toward eliminating pregnancy complications related to these conditions.

Ethical Considerations

Compliance with ethical guidelines

Permission was obtained from the Jordanian Research Ethics Committee at the Ministry of Health (No.: 12264).

Funding

This research received no specific grant from funding agencies in the public, commercial, or not-for-profit sectors.

Authors' contributions

Conceptualization, data analysis and writing the drafting the manuscript: Lourance Al Hadid and Omymah Z. Al-Rajabi; Data conduction, curation: Marwa Al Barmawi and Omymah Z. Al-Rajabi; Writing editing & review: Ahmad Yahya AL-Sagarat and Marwa Al Barmawi; Final approval: All authors.

Conflict of interest

The authors have declared no conflict of interest.

Acknowledgments

We thank all women who participated in this study. We appreciate all nurses who facilitated data collection from women. We would also like to say that several women were sent for treatment of their condition after they were informed about the results of their tests in this study.

References

- World Health Organization (WHO). Global database on body mass index: BMI classification [Internet]. 2006 [Updated 2023 June 11]. Available from: [Link]
- [2] Driscoll AK, Gregory ECW. Increases in prepregnancy obesity: United States, 2016-2019. NCHS Data Brief. 2020; (392):1-8. [PMID]
- [3] Chopra M, Kaur N, Singh KD, Maria Jacob C, Divakar H, Babu GR, et al. Population estimates, consequences, and risk factors of obesity among pregnant and postpartum women in India: Results from a national survey and policy recommendations. International Journal of Gynaecology and Obstetrics. 2020; 151(Suppl 1):57-67. [DOI:10.1002/ijgo.13319] [PMID] [PMCID]
- [4] Langley-Evans SC, Pearce J, Ellis S. Overweight, obesity and excessive weight gain in pregnancy as risk factors for adverse pregnancy outcomes: A narrative review. Journal of Human Nutrition and Dietetics. 2022; 35(2):250-64. [DOI:10.1111/jhn.12999] [PMID] [PM-CID]
- [5] Cirulli F, Musillo C, Berry A. Maternal obesity as a risk factor for brain development and mental health in the offspring. Neuroscience. 2020; 447:122-35. [DOI:10.1016/j.neuroscience.2020.01.023]
 [PMID]
- [6] Nalbant-Poyraz E, Farebrother J. Overweight and obesity may be a risk factor for thyroid dysfunction in pregnant women: A systematic review and meta-analysis. Proceedings of the Nutrition Society. 2021; 80(OCE5):E199. [DOI:10.1017/S002966512100327X]
- [7] Wei W, Zhang X, Zhou B, Ge B, Tian J, Chen J. Effects of female obesity on conception, pregnancy and the health of offspring. Frontiers in Endocrinology. 2022; 13:949228. [DOI:10.3389/fendo.2022.949228] [PMID] [PMCID]
- [8] Costantine MM, Smith K, Thom EA, Casey BM, Peaceman AM, Varner MW, et al. Effect of thyroxine therapy on depressive symptoms among women with subclinical hypothyroidism. Obstetrics and Gynecology. 2020; 135(4):812-20. [DOI:10.1097/ AOG.000000000003724] [PMID] [PMID]
- [9] Tang Y, Chen Y, Feng H, Zhu C, Tong M, Chen Q. Is body mass index associated with irregular menstruation: A questionnaire study? BMC Womens Health. 2020; 20(1):226. [DOI:10.1186/s12905-020-01085-4] [PMID] [PMCID]
- [10] Gambineri A, Laudisio D, Marocco C, Radellini S, Colao A, Savastano S, et al. Female infertility: Which role for obesity? International Journal of Obesity Supplements. 2019; 9(1):65-72. [DOI:10.1038/ s41367-019-0009-1] [PMID] [PMCID]
- [11] Casas M, Chatzi L, Carsin AE, Amiano P, Guxens M, Kogevinas M, et al. Maternal pre-pregnancy overweight and obesity, and child neuropsychological development: Two Southern European birth cohort studies. International Journal of Epidemiology. 2013; 42(2):506-17. [DOI:10.1093/ije/dyt002] [PMID]
- [12] Poppe K, Bisschop P, Fugazzola L, Minziori G, Unuane D, Weghofer A. 2021 European thyroid association guideline on thyroid disorders prior to and during assisted reproduction. European Thyroid Journal. 2021; 9(6):281-95. [DOI:10.1159/000512790] [PMID] [PMCID]
- [13] Dong AC, Stagnaro-Green A. Differences in diagnostic criteria mask the true prevalence of thyroid disease in pregnancy: A systematic review and meta-analysis. Thyroid. 2019; 29(2):278-89. [DOI:10.1089/thy.2018.0475] [PMID]

- [14] Robillard PY, Dekker G, Scioscia M, Bonsante F, Iacobelli S, Boukerrou M, et al. Increased BMI has a linear association with lateonset preeclampsia: A population-based study. Plos One. 2019; 14(10):e0223888. [DOI:10.1371/journal.pone.0223888] [PMID] [PMCID]
- [15] Krönke AA, Jurkutat A, Schlingmann M, Poulain T, Nüchter M, Hilbert A, et al. Persistent organic pollutants in pregnant women potentially affect child development and thyroid hormone status. Pediatric Research. 2022; 91(3):690-8. [DOI:10.1038/s41390-021-01488-5] [PMID] [PMCID]
- [16] Gárate-Escamilla AK, Garza-Padilla E, Carvajal Rivera A, Salas-Castro C, Andrès E, Hajjam El Hassani A. Cluster analysis: A new approach for identification of underlying risk factors and demographic features of first trimester pregnancy women. J Clin Med. 2020; 9(7):2247. [DOI:10.3390/jcm9072247] [PMID] [PMCID]
- [17] Toloza FJK, Abedzadeh-Anaraki S, Maraka S. Subclinical hypothyroidism in pregnancy. Current Opinion in Endocrinology, Diabetes, and Obesity. 2019; 26(5):225-31. [DOI:10.1097/ MED.00000000000491] [PMID]
- [18] Engel SM, Villanger GD, Herring A, Nethery RC, Drover SSM, Zoeller RT, et al. Gestational thyroid hormone concentrations and risk of attention-deficit hyperactivity disorder in the Norwegian Mother, Father and Child Cohort Study. Paediatric and Perinatal Epidemiology. 2023; 37(3):218-28. [DOI:10.1111/ppe.12941] [PMID]
- [19] Department of Statistic. Population in Jordan. Department of Statistic. Amman: 2015. [Link]
- [20] Faul F, Erdfelder E, Buchner A, Lang AG. Statistical power analyses using G*Power 3.1: Tests for correlation and regression analyses. Behavior Research Methods. 2009; 41(4):1149-60. [DOI:10.3758/ BRM.41.4.1149] [PMID]
- [21] Wright CM, Cole TJ, Fewtrell M, Williams JE, Eaton S, Wells JC. Body composition data show that high BMI centiles overdiagnose obesity in children aged under 6 years. The American Journal of Clinical Nutrition. 2022; 116(1):122-31. [DOI:10.1093/ajcn/ nqab421] [PMID] [PMCID]
- [22] Cox JL, Holden JM, Sagovsky R. Detection of postnatal depression. development of the 10-item edinburgh postnatal depression scale. The British Journal of Psychiatry. 1987; 150:782-6. [DOI:10.1192/ bjp.150.6.782] [PMID]
- [23] Knight-Agarwal CR, Jani R, Al Foraih M, Eckley D, Lui CKW, Somerset S, et al. Maternal body mass index and country of birth in relation to the adverse outcomes of large for gestational age and gestational diabetes mellitus in a retrospective cohort of Australian pregnant women. BMC Pregnancy and Childbirth. 2021; 21(1):649. [DOI:10.1186/s12884-021-04125-5] [PMID] [PMCID]
- [24] Vitner D, Harris K, Maxwell C, Farine D. Obesity in pregnancy: A comparison of four national guidelines. The Journal of Maternal-Fetal & Neonatal Medicine. 2019; 32(15):2580-90. [DOI:10.1080/1 4767058.2018.1440546] [PMID]
- [25] Menon S, Sivaprasad K. Maternal and neonatal outcomes in pregnancies complicated by maternal obesity. International Journal of Reproduction, Contraception, Obstetrics and Gynecology. 2019; 8(2):474-9. [DOI:10.18203/2320-1770.ijrcog20190270]
- [26] Hanif S, Zubair M, Shabir N, Zia MS. A comparative study of maternal and fetal outcome in obese and non-obese pregnant women. Journal of The Society of Obstetricians and Gynaecologists of Pakistan. 2020; 10(2):96-101. [Link]



- [27] Lee SY, Pearce EN. Testing, monitoring, and treatment of thyroid dysfunction in pregnancy. The Journal of Clinical Endocrinology and Metabolism. 2021; 106(3):883-92. [DOI:10.1210/clinem/dgaa945] [PMID] [PMCID]
- [28] Stubert J, Reister F, Hartmann S, Janni W. The risks associated with obesity in pregnancy. Deutsches Arzteblatt International. 2018; 115(16):276-83. [DOI:10.3238/arztebl.2018.0276] [PMID] [PMCID]
- [29] Pillai NS, Bennett J. Prevalence of hypothyroidism amongst pregnant women: A study done in rural set up. International Journal of Reproduction, Contraception, Obstetrics and Gynecology. 2018; 7(4):1586-92. [DOI:10.18203/2320-1770.ijrcog20181360]
- [30] Mammen JSR, Cappola AR. Autoimmune thyroid disease in women. JAMA. 2021; 325(23):2392-3. [DOI:10.1001/jama.2020.22196]
 [PMID] [PMID]
- [31] Alkharaiji M, Anyanwagu U, Donnelly R, Idris I. Tier 3 specialist weight management service and pre-bariatric multicomponent weight management programmes for adults with obesity living in the UK: A systematic review. Endocrinology, Diabetes & Metabolism. 2018; 2(1):e00042. [DOI:10.1002/edm2.42] [PMID] [PMCID]
- [32] Yehia DBM, Malak MZ, Al-Thwabih NN, Awad RR, Al-Ajouri ES, Darwish SS, et al. Psychosocial factors correlate with fatigue among pregnant women in Jordan. Perspectives in Psychiatric Care. 2020; 56(1):46-53. [DOI:10.1111/ppc.12372] [PMID]